Author's response to reviews

Title: Economic burden of stroke in a large county in Sweden

Authors:

Josefine Persson (josefine.persson@vgregion.se)
José Ferraz-Nunes (ferraz@bostream.nu)
Ingvar Karlberg (ingvar.karlberg@gu.se)

Version: 2 Date: 20 December 2011

Author's response to reviews: see over
BMC Health Service Research

Submission of manuscript to BMC Health Service Research

We hereby wish to submit enclosed manuscript of “Economic burden of stroke in a large county in Sweden”. Our manuscript provides detailed information covering every part throughout the whole chain of health care in addition to societal costs for stroke patients in western Sweden. Thereby, the result in this study can be used as a basis to evaluate new improvements as well as to initiate a discussion of cost and priorities.

Changes which have been made:
- Patients and method – Data source: information regarding datasource have been updated
- Competing interest: Details regarding competing interest have been added
- Authors contributions: Details regarding authors contributions have been added

Josefine Persson, MSc
Sahlgrenska Academy, Gothenburg University
Box 400
405 30 Gothenburg, Sweden
Telephone: +46 739 547446
Email: josefine.persson@vgregion.se

José Ferraz-Nunes
Senior Lecturer
University West
461 86, Trollhättan, Sweden
Telephone: +46 703354676
Email: ferraz@bostream.nu